Author’s response to reviews

Title: Colorectal endoscopic mucosal resection with submucosal injection of epinephrine versus hypertonic saline in patients taking antithrombotic agents: propensity-score-matching analysis

Authors:

Daisuke Yamaguchi (daisukehawks@gmail.com)
Hisako Yoshida (yoshida.hisako@med.osaka-cu.ac.jp)
Kei Ikeda (ikeda.kei.zb@mail.hosp.go.jp)
Yuki Takeuchi (sngjw899@yahoo.co.jp)
Shota Yamashita (sho.10gl29@gmail.com)
Amane Jubashi (ajubashi1989@gmail.com)
Takahiro Yukimoto (koitarou1800@gmail.com)
Eri Takeshita (eltsyn523@gmail.com)
Wataru Yoshioka (sailingxyz94@yahoo.co.jp)
Hiroko Fukuda (hirofukuda_985@yahoo.co.jp)
Naoyuki Tominaga (tominaga-n@koseikan.jp)
Nanae Tsuruoka (nana.tsuruoka@gmail.com)
Tomohito Morisaki (t_msaki@hotmail.co.jp)
Keisuke Ario (ario.keisuke.kx@mail.hosp.go.jp)
Seiji Tsunada (tsunada.seiji.ed@mail.hosp.go.jp)
Kazuma Fujimoto (fujimoto.kazuma@iuhw.ac.jp)

Version: 3  Date: 07 Nov 2019

Author’s response to reviews:

1. As indicated in our submission guidelines, please include the following sections in your manuscript: Background, Methods, Results, Discussion, Conclusions.
I appreciate the favorable comment raised by the reviewer. According to the comment, I have revised the text.

2. Please include an Acknowledgments subsection in your Declarations. The Acknowledgments subsection is your opportunity to acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials. If you have no individuals to acknowledge you may write “Not applicable” in this subsection.

Accordingly, I have added an Acknowledgments subsection.

3. If two authors have the same initials, please distinguish between them by designating them with numbers 1 and 2. For example: TY1 and TY2. TY1 would correspond to the author furthest up on the author list.

I have distinguished between the same initials.

4. Please provide more justification for the contributions of authors. We ask that you consider your list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Currently, the contributions of authors YT, KI, TY1, SY, AJ, TY2, ET, WY, HF, NT1, NT2, TM, KA, and ST do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. Authors are expected to fulfil the criteria below (adapted from McNutt et al., Proceedings of the National Academy of Sciences, Feb 2018, 201715374; DOI: 10.1073/pnas.1715374115; licensed under CC BY 4.0):

Each author is expected to have made substantial contributions to the conception OR design of the work; OR the acquisition, analysis, OR interpretation of data; OR the creation of new software used in the work; OR have drafted the work or substantively revised it

AND to have approved the submitted version (and any substantially modified version that involves the author's contribution to the study);

AND to have agreed both to be personally accountable for the author's own contributions and to ensure that questions related to the accuracy or integrity of any part of the work, even ones in which the author was not personally involved, are appropriately investigated, resolved, and the resolution documented in the literature.
Acquisition of funding, collection of data or general supervision of the research group, alone, does not usually justify authorship.

Anyone listed as an author must be included in this section. If you choose to change your author list you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found at this web address https://resource-cms.springernature.com/springer-cms/rest/v1/content/7454878/data/v5. All instructions can be found on the form. Please return the form within 14 days with all author signatures.

Anyone who contributed towards the article who does not meet the criteria for authorship can be acknowledged in the ‘Acknowledgements’ section.

I have provided more justification for the contributions of authors.

According to the comment, I have revised the Authors’ contributions.

DY and KF designed the research; DY, YT, KI, TY1, SY, AJ, TY2, ET, WY, HF, NT1, NT2, TM, KA and ST performed colorectal endoscopic mucosal resection procedure, acquired the patient data, and made substantial contributions to the conception; DY designed the work; DY and HY analyzed the data; DY and KF wrote the paper; KF was a major contributor in writing the manuscript. All authors read, have approved the submitted version and approved the final manuscript.

All authors have agreed both to be personally accountable for the author's own contributions and to ensure that questions related to the accuracy or integrity of any part of the work, even ones in which the author was not personally involved, are appropriately investigated, resolved, and the resolution documented in the literature.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

According to the comment, I have uploaded my manuscript.